Author's response to reviews

Title: The relation between celiac disease, nonceliac gluten sensitivity and irritable bowel syndrome

Authors:

Magdy El-Salhy (magdy.el-salhy@helse-fonna.no)
Jan Gunnar Hatlebakk (jan.hatlebakk@helse-bergen.no)
Odd Helge Gilja (odd.helge.gilja@helse-bergen.no)
Trygve Hausken (trygve.hausken@helse-bergen.no)

Version: 3 Date: 30 July 2015

Author's response to reviews:

Reviewer 1: Imran Aziz
1. Changes in the paragraph on IBS and CD was made as recommended by the reviewer (page 5, second paragraph).
2. The paragraph on is gluten responsible for NCGS? Was rephrased as recommended by the reviewer (page 7, second paragraph).
3. The paragraph on is gluten responsible for NCGS? Was revised as recommended by the reviewer (page 7, second paragraph).
4. The paragraph on is gluten responsible for NCGS? The possible role of amylase-trypsin inhibitors was added as recommended by the reviewer (page 7, third paragraph).
5. Compelling evidence was replaced by evidence throughout the manuscript as recommended by the reviewer

Reviewer 2: Adriana Franzese
1. Table 1 and 2 are correctly placed in the manuscript as recommended by the reviewer (pages 3 and 5).
2. The endocrine cells were described in the legends.
3. All the facts referred to in the manuscript was included, and one can not refer to facts presented before 2000 without mentioning the sources.